Author's response to reviews

Title: A conceptual model for worksite intelligent physical exercise training - IPET - intervention for decreasing life style health risk indicators among employees: A randomized controlled trial.

Authors:

Gisela Sjogaard (gsjogaard@health.sdu.dk)
Just B Justesen (ibijustesen@health.sdu.dk)
Mike Murray (mmurray@health.sdu.dk)
Tina Dalager (tidal08@student.sdu.dk)
Karen Sogaard (ksogaard@health.sdu.dk)

Version: 2 Date: 2 December 2013

Author's response to reviews:

Ethical and Funding Approval Documentation: The study has received private funding as stated in the acknowledgement. Additionally, the University of Southern Denmark has funded the project in terms of salary for a PhD student. In this context the study protocol was assessed by the university’s Project committee for the dean, where external evaluators have been involved. The first version was revised and finally approved. Letter of approval and funding is send to Journal Editorial Office at BMCSeriesEditorial@biomedcentral.com.

Status: This is an ‘ongoing’ study, where the investigators are still collecting data.

Related articles: I confirm that no results from this study have been published or submitted elsewhere.